Author's response to reviews

**Title:** The CAP study, evaluation of integrated universal and selective prevention strategies for youth alcohol misuse: Study protocol of a cluster randomized controlled trial

**Authors:**

Nicola C Newton (nickien@unsw.edu.au)
Maree Teesson (m.teesson@unsw.edu.au)
Emma L Barrett (e.barrett@unsw.edu.au)
Tim Slade (tims@unsw.edu.au)
Patricia J Conrod (patricia.conrod@umontreal.ca)

**Version:** 3  **Date:** 16 August 2012

**Author's response to reviews:**

Dear Deesha,

Our revised manuscript has now been uploaded which no longer includes the additional files (proof of ethics and funding).

Best wishes,

Nicola Newton.